Author’s response to reviews

Title: Designing a customized clinical practice guideline regarding antibiotic prophylaxis for Iranian general dentists

Authors:

Najmeh Savadi (savadin@sums.ac.ir; savadi_naj@yahoo.com)
Omid Barati (omidbarati40@gmail.com)
Hossein Mirhadi (Mirhadid@sums.ac.ir)
Ali Golkari (aligolkari@yahoo.com)

Version: 2 Date: 23 Aug 2019

Author’s response to reviews:

Dear BMC Oral Health Editorial Office

The authors would like to thank you and the reviewers for instructive and motivative comments.

Below is the point to point response to reviewer’s comments:

1. The manuscript is now edited by a native English speaking colleague and some minor corrections have been made.

2. Minor changes in the subtitles of method section has been made including:
   
   I. Line 133, page 6
   
   II. Line 139, page 6
   
   III. Line 180, page 8

Kind regards,

Najmeh Savadi,

Corresponding author